Author's response to reviews

Title: Is there an association between low dose aspirin and anemia (without overt bleeding)?: narrative review

Authors:

Helen Gaskell (helen.gaskell@pru.ox.ac.uk)
Sheena Derry (sheena.derry@pru.ox.ac.uk)
Andrew Moore (andrew.moore@pru.ox.ac.uk)

Version: 3 Date: 28 September 2010

Author's response to reviews:

No changes to this version of the manuscript were requested by the referees.

We have made minor changes to the manuscript as requested by the editors (e.g., removing highlighting which had been used to clarify where there had been changes in the manuscript following review of the first version of the manuscript by the referees). We updated the authors' addresses, corrected two typographical errors, and modified the wording of the paragraph on the authors' contributions, as requested by the editors.

In addition, we have added a "note in proof", which is self-explanatory.